Author's response to reviews

Title: Tobacco use induces anti-apoptotic, proliferative patterns of gene expression in circulating leukocytes of Caucasian males

Authors:

Peter C Charles (pcharles@med.unc.edu)
Brian D Alder (bdaldger@gmail.com)
Eleanor G Hilliard (eleanorh@med.unc.edu)
Jonathan C Schisler (jonathan_schisler@med.unc.edu)
Robert E Lineberger (rob_lineberger@med.unc.edu)
Joel S Parker (jparker@expressionanalysis.com)
Sabeen Mapara (sabeen_mapara@gmail.com)
Samuel S Wu (ssw@med.unc.edu)
Andrea Portbury (andrea_portbury@med.unc.edu)
Cam Patterson (cpatters@med.unc.edu)
George A Stouffer (rick_stouffer@med.unc.edu)

Version: 3 Date: 27 June 2008

Author's response to reviews: see over
Dear Sirs-

We have made the requested formatting changes to our manuscript, and corrected the remaining typographical errors that we were able to find. A point-by-point response to your editorial suggestions can be found below.

**Formatting changes requested**

-------------------------------

Title - Please use sentence case in the title (capitalise only the first word, and proper nouns), both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

>Only the first word of the title, and the proper noun “Caucasian” are capitalized.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

>Author list matches submission system author list.

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names. Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers. Please write each affiliation on a new line.

>Authors’ affiliations have been corrected to conform to your requirements.

‘Transcriptional Signals of Tobacco Use’
Please remove this text from the head of each page.

>Running title has been removed from the page headings.

Website addresses - Websites must be removed from the text, and cited from the main reference list only. If you are using referencing software such as EndNote or Refman, please ensure that your references are entered through the software rather than included in the manuscript reference list manually. Please present your reference in the following structure, e.g.


>All references to websites have been included in the reference list.

Endnote - There is a discrepancy between the number of references in your manuscript reference list, and the number extracted by our system. This can be seen on the upload page of the submission system in the main manuscript box. For help please visit: http://www.biomedcentral.com/info/ifora/endnote/#technicalhelp
>We were unable to format our references in EndNote such that your system would recognize them. Because of the repeated failures, and our inability to get more that 24 references recognized, the document has been reformatted as an RTF, and submitted in this fashion. We understand that this will incur some additional editorial costs.

Please re-title 'Figures' 'Figure legends'.

>Figure legends have been re-titled.

Table with roman numerals - Please do not number the tables using Roman Numerals. Please rename all tables using standard numbers e.g. Table 1, Table 2 etc. Please also remember the change your references to the tables throughout the manuscript text.

>Tables have been re-numbered with Arabic numerals.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

> All figures have been cropped to the smallest size that affords acceptable viewing.

Please feel free to contact me via email (pcharles@med.unc.edu) or telephone (001.919.843.1610) if you require any additional changes.

Best regards,

Peter C. Charles, PhD